Math 221 Review sheet for Test 1 Spring 2001

1. NO CALCULATORS ALLOWED 2. Test 1 covers the Chapter 0 handout, Chapter 1 from the text and the determinates handout. Anything done in class is fair game for the test.

3. There are ten problems worth a total of 100 points. 4. There are two problems from the pretest, worth 5 points each. 5. Theory: 3 questions, 20 points total. Know basic definitions. Know properties of matrix and vector algebra and of determinates. Be able to do short proofs from the handouts, the text, class notes, and your homework.

6. The rest of the test (70 points) involves calculations. Of course understanding the theory can make some calculations easier.